Author's response to reviews

Title: Surmounting difficulties to provide home based neonatal care - reflections of community health workers

Authors:

Abhijit Prabhughate (abhijit.prabhughate@ambujacement.com)
Pearl Tiwari (pearl.tiwari@ambujacement.com)
Shubhangi Sohoni (shubhangi.sohoni.ext@ambujacement.com)
Vallaree Morgaonkar (vallarimorgaonkar@gmail.com)
Ajay Phatak (ajaygp@gmail.com)
Somashekhar Nimbalkar (somu_somu@yahoo.com)
Anangha Mahajani (anagha.mahajani.ext@ambujacement.com)

Version: 1 Date: 06 Jul 2017

Author's response to reviews:

1. Please add a short, brief Conclusion section after the Discussion section of your main manuscript, in order to adhere to journal submission requirements.

Ans: We have added a heading Conclusion appropriately before the last paragraph.

2. Thank you for providing a section on Ethics approval. We would appreciate it if you could please provide us with the full, official name of the committee which granted you approval for this study, and committee reference number if appropriate.

Ans: We have given the name. There is no document number or committee number that can be given.
3. Thank you for providing a section on author contribution. We use this information to note which author contributed what to the study. We had noticed only six of the seven authors are mentioned in this section. Please revise this section to include all authors, so that all authors listed in the manuscript can qualify for authorship.

Ans: We have listed all authors in the initial document. Ajay was referred to by name and not by a initial. He is now referred to as AGP.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We respectfully request you please amend this statement to adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

If you would please amend the statement to reflex these requirements, we would appreciate it.
Additionally, please use this statement to confirm that all authors have read and approved of the final manuscript.

4. We would appreciate if you added a section titled “Supplementary files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

Ans: We have added a section as Supplementary File

Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

5. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.

Ans: Included in main manuscript